

Author's response to reviews

Title: A study of HIV/AIDS related knowledge, attitude and behaviors among female sex workers in Shanghai China

Authors:

Hong Huang (huanghong096@hotmail.com)

Yong Cai (caiyong202028@163.com)

Tian Shen (pedy@yahoo.cn)

Bei Pei (babystar009@yahoo.com.cn)

Xueqin Jiang (jiangxueqin@sjtu.edu.cn)

Rong Shi (shirong61@yahoo.com.cn)

Shenghui Li (Lsh9907@163.com)

Gang Xu (smile567@sohu.com)

Xiuxia Ye (xiuxiaye@gmail.com)

Meili Shang (lilyshang1984@163.com)

Version: 4 **Date:** 21 May 2010

Author's response to reviews: see over

Dear Prof. Nina Titmus, the Scientific Editor:

We understand that BMC-series Journals are high profile journals. We have re-revised our manuscript carefully again according to Professor Juan Pablo Gutierrez's comments.

Thank you for your reconsideration.

Best regards,

Hong Huang

Point to Point Reply To Referee 1: Reviewer: Juan Pablo Gutierrez

Thank you very much for your suggestion to our manuscript and we do learn much from you and your suggestion which can really promote the quality of our manuscript.

1. We have added the references on the ICC value used to adjust sample in the revised manuscript.
2. We check the figure in Table 1 and find a couple of p-values greater than 1, such as 1.067 and 14.432, this two value and 0.473,0.751 are not P-values, they are all value of χ^2_{rs} adjusted from Pearson chi-square. It's our mistake that not explain it clearly in the table list and we have revised it. Thank you for your reminding.
3. High quality of written English is very important for publication, and we have tried our best to improve it by revising it again and again, the language corrections are needed since we are from non-English speaking country. Thank you very much.